Author’s response to reviews

Title: Case Presentation: Implantation of Cardiac Resynchronization Therapy Pacemaker via the Coronary Sinus in a Patient with Triple Valve Replacement

Authors:

cheng zheng (793601136@qq.com)
Weiqian Lin (373806983@qq.com)
Yuanzheng Lin (408811608@qq.com)
Hao Lian (184900642@qq.com)
Zhirui Liu (410034577@qq.com)
Jiahui Chen (jiahuits@qq.com)
jia-feng Lin (linjiafeng_wzmcfey@163.com)

Version: 4 Date: 08 Feb 2018

Author’s response to reviews:

Dear editor,

We really thanks for your suggestion to improve our paper. We have done revision according to your suggestion and included a cover letter with a point-by-point response to your comments.

Point-by point response:

1. Please remove the statement “the patient provided written, informed consent before the procedure and gave consent for her information to be used for this study.” From the section ‘Ethics approval and consent to participate’ as the patient underwent standard care consent to participate was not required.

Answer: Dear editor, we have removed the statement “the patient provided written, informed consent before the procedure and gave consent for her information to be used for this study.” from the section ‘Ethics approval and consent to participate’.

2. We note that you have not included a ‘Consent for publication’ section in Declarations. If identifying images or other personal or clinical details of participants are presented (as in a case report) that compromise anonymity, a statement of consent to publish from the
patient should be included. This section must be included even if it is not applicable to your manuscript. Please indicate whether the patient gave consent for the publication of clinical data/figures and how consent was given.

Answer: Dear editor, based on your requirement, we asked the patient for the permission of publication. The patients agreed and signed the consent for publication. We have provided the statement of consent to publish in the supplementary file. Meanwhile, in the Consent for publication’ section in Declarations, we made a declaration as follow: Written informed consent was obtained from the patient for publication of this case report and any accompanying images and tables. A copy of the written consent is available for review by the Editor of this journal.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: Dear editor, we have uploaded our manuscript, tables, figures and supplementary files according to your request.

Thanks sincerely,

Jia-feng.